CoRIPS Research Award 089 Beverley Atherton Awarded £3569

Can the early signs and symptoms suggestive of spinal cord compression be identified by radiographers during bone scans by gathering clinical information about back pain from patients?

Lay summary

The study aims to find out if radiographers who perform bone scans can gather information from patients about back pain which may identify early signs and symptoms suggestive of Metastatic Spinal Cord Compression (MSCC). This is not currently routine practice and there is no evidence that radiographers will be able to successfully collect this sort of clinical information. If data can successfully be obtained it may be possible to use it in conjunction with other clinical information to identify patients at high risk of developing MSCC. These patients may benefit from prompt referral for medical assessment or Magnetic Resonance Imaging (MRI) scanning.

The project will assess back pain using two questionnaires in 200 patients or over four months, whichever is reached first. An existing, validated pain questionnaire, the Brief Pain Inventory (BPI) will be used as well as a new questionnaire which has been developed for use in this study. The new questionnaire is based on the early signs and symptoms of spinal pain and neurological symptoms as defined by the National Institute for Health and Clinical Excellence (NICE). The data from the questionnaires and other clinical information acquired during the bone scan visit will be compared with clinical follow up of the patients 5 months after their visit, to see if any have required treatment for malignant spinal disease. Each element of the collected data will be compared to see if any relationship exists. The project is the first stage in exploring the possibility of developing a risk assessment tool for use in an imaging setting.

Description of the project:

Principal aim of the study

The study aims to find out if radiographers involved in performing radioisotope bone scans can gather clinical data from patients regarding back pain which may identify early signs and symptoms suggestive of Metastatic Spinal Cord Compression (MSCC). Currently this is not routine practice and there is no evidence that radiographers will be able to successfully do this. If data can be collected it may be possible to use it in conjunction with standard clinical data to identify those patients at high risk of developing MSCC and those who may benefit from prompt referral for medical assessment or Magnetic Resonance Imaging (MRI).

Primary research question

Is it feasible and acceptable for radiographers to gather clinical information from patients regarding back pain during a bone scan procedure which may be indicative of metastatic spinal disease using i) a new questionnaire which has been designed, based on NICE indications and ii) a standardised pain questionnaire, the Brief Pain Inventory (BPI)?

Secondary research questions

1. Is there a relationship between data collected using the BPI and that collected using NICE signs and symptoms questionnaire?

2. Is there any relationship between questionnaire data, standard clinical data acquired during a routine bone scan procedure and clinical outcome in terms of requirement for treatment of metastatic spinal disease?

Outcomes / Deliverables

The study will determine if data can be collected which may provide evidence of the early signs and symptoms of MSCC in a group of patients who may all be at potential risk. Based on the outcome of this project, a wider future study may provide scoping information regarding the feasibility of developing and validating a risk assessment tool. This could potentially be used routinely in clinical practice to identify high risk patients at an early and opportune time. By grouping risk tool scores it may be possible to signpost patients at an early stage to an appropriate interventions thus potentially sparing them from developing disabling clinical complications. The study fulfils the requirements of a feasibility study by evaluating acceptability and practicality (Bowen et al 2009). A risk assessment tool could have particular value in settings where oncology patients are infrequent and the knowledge of early signs of MSCC are less well established.

Review of the literature and identification of the current gap in knowledge

What is MSCC?

MSCC is an oncological emergency which occurs in some patients with cancer. NICE defines it as pathological vertebral body collapse or direct tumour growth causing compression of the spinal cord or cauda equina. Treatment includes radiotherapy and/or surgery.

Why is it important?

Without timely treatment MSCC can cause irreversible neurological damage, permanent disability or premature death. Early diagnosis and treatment makes a significant difference to quality of life but there is substantial evidence that late presentation is not uncommon (Husband 1998). The neurological status at the time of diagnosis of MSCC is a significant predictor of outcome in terms of survival (Husband 2001, Levack et al 2002). Evidence from several seminal observational studies indicates that once MSCC is suspected, it is essential that investigation, planning and treatment takes place before any further loss of neurological function (Helweg-Larsen et al 1996, Husband et al.1998, Levack et al. 2002, Maranzano et al. 1995, Solberg et al.1999, Turner et al. 1993).

What is the size of the problem?

An estimated 3-5% of all patients with cancer will develop spinal metastases (NICE 2008), most commonly in cases of breast, prostate or lung cancer, in whom the incidence may be as high as 19 %. These high risk cancers account for over 50% of cases of MSCC (Loblaw et al 2003, Levack et al 2002). However, 23% of patients with MSCC have no prior cancer diagnosis (NICE 2008) and the onset of neurological symptoms is often insidious (IAEA 2007). NICE issued guidance in Nov 2008 regarding the diagnosis and management of patients at risk of or with MSCC followed by regional guidance in Sept 2010.

What are the key signs, symptoms and risk factors?

Back pain is the most frequent first symptom of MSCC, occurring in 95% of patients (Levack et al 2002) but it is also a common complaint in the general population. NICE have identified a specific list of features of back pain and neurological symptoms that are better predictors of MSCC. They recommend that health care professionals should be aware of the signs and symptoms of MSCC but acknowledge that those which are taught often relate to later neurological stages of the condition (Bucholtz 1999). Regional guidance states that the professional who recognises that cord compression is a potential complication should undertake to make patients aware of potential signs (MCCN 2012). NICE suggest that standardised written information is given to patients identified as being at risk about the symptoms and details of what they, or their families/carers, should do if they develop them. An agreed plan to disseminate such information has yet to be implemented and anecdotal evidence suggests that clinicians have difficulty in reaching an agreement as to which patients and at which stage of treatment patients should recieve risk information as it may cause an unnecessary burden of concern.

The role of imaging - an opportunity

Radioisotope bone scanning is not used to diagnose MSCC but is a routinely performed examination on the same group of patients who may potentially be at risk of developing the condition. The visit to the hospital for the bone scan therefore, presents an opportunity to assess if patients are demonstrating any early symptoms. Bone scans are commonly used as a routine investigation to assess the whole body for bony metastases. The procedure does not, however, demonstrate the extent of soft tissue compression or reliably detect the level of a MSCC. Locally, a recent audit showed that 34% of bone scans performed on patients with known or suspected cancer demonstrated metastases, with 28% having spinal metastases (83% of cases with metastases had spinal metastases). MRI is the imaging modality of choice for diagnosing MSCC (Cook 1998) and is central to NICE guidance to the diagnosis, staging and planning of treatment. Regular MRI imaging of the spine to detect early disease is not, however, recommended if patients are asymptomatic (NICE 2008). It is not uncommon that patients having MRI scans for suspected of MSCC may have previously undergone a radioisotope bone scan. A recent audit within this unit indicated this was the case in almost 62% of patients. 26% of these patients had bone scans within the previous month, 48% within the previous six months and 65% within the previous year. The actual number may be higher as bone scans may have also been performed at hospitals whose records are not included in the local shared computer system.

Bone scans referrals may be received from a variety of referral routes, oncology consultants, the local district general hospital, some GPs and private practice. An audit indicated that 96% of the total referrals to the department had a known or suspected cancer (91% with a known cancer, 5% with a suspected cancer). Of these with known or suspected cancers, 68% were referred from oncology consultants and 32% from outside the oncology setting. Of the total number of cases with known or suspected Cancer 85% had known primaries defined as being high risk of developing metastatic spinal disease (NICE).

The request to perform a bone scan rather than MRI implies that the clinician was not suspicious of MSCC at the time of referral. Symptoms may, however, have altered by the time the patient attends for their appointment or the referral may be from a clinician whose primary specialism is outside oncology and the risk of MSCC may not have been promptly considered.

Method

The study is of quantitative, descriptive and exploratory cohort design. Patients will be asked to complete two questionnaires during their visit for a nuclear medicine bone scan.

Sample - Patients will be recruited from those referred for whole body bone scanning who have a known history or suspicion of cancer.

Inclusion criteria

- Patients referred for whole body bone scans with a known or suspected cancer.
- Over eighteen years of age.
- Able to communicate in English.
- Capable of giving written informed consent.

Exclusion

- Known existing or previous MSCC
- Patients not having been given appropriate time to consider the study
- Patients who have already participated in the study. If patients attend for a second bone scan whilst the study is being conducted they will be ineligible to participate.

Data will be collected from a 200 patients or over a period of four months, whichever occurs sooner. Approximately 75 patients per month with a known or suspected cancer are currently referred for bone scans. If 50 (66%) of eligible patients per month can be recruited then 200 patients can be entered in the study over approximately four months.

Recruitment

The project will be led by the Diagnostic Research Radiographer who together with the Nuclear Medicine Clinical Specialist will identify eligible patients from referral forms. An invitation to participate in the study and a patient information leaflet will be posted to the patient with their appointment letter. Patients will be given a telephone number to contact staff should they require further information. Approximately five whole time equivalent staff will be available to participate in the study. Patients will be consented by qualified, trained staff on arrival in the department before the clinical procedure.

Staff

Experienced oncology, nuclear medicine radiographers with Good Clinical Practice research (GCP) and research project training will undergo a short education training session (approximately one/two hours). The training will be provided by the local clinical lead for MSCC management. This will include information and training regarding:

- Completion of the BPI
- The NICE related back pain questionnaire which has been developed
- NICE and regional MSCC guidance
- Clarifying the early clinical symptoms suggestive of spinal metastases and the neurological symptoms suggestive of MSCC.

Data collection

Patients will be asked to complete two questionnaires during their visit.

i) NICE based back pain questionnaire -this will be completed by the radiographer during the routine patient interview conducted prior to the bone scan injection. The patient is asked a specific set of questions about back pain based on NICE indications. The responses will be recorded as "yes" or "no" with an opportunity to comment when a definitive response cannot be established. Both radiographers and patients are then asked to score the questionnaire for ease/difficulty of completion using a linear 1-10 scale and a free text area is also available for either to comment in.

ii) BPI- A bone scan involves a wait of three hours between injection and imaging. Patients will be asked to complete the BPI independently during this waiting time but assistance will be offered and available if required.

iii) Clinical Data - Clinical data will be recorded by the radiographer. This will include:

- Site of primary cancer if known / suspicion of cancer
- Whether back pain is mentioned on the referral form
- Radiographer opinion on reviewing the scan images (presence of metastases, presence of spinal metastases)

The radiologist report will be noted.

On line case notes and data from the regional MSCC co-ordinator will be reviewed after five months to assess if any patients have required treatment for metastatic spinal disease.

Data analysis

The Research Radiographer will perform all the following data analysis.

i) Analysis of Primary aims

a) Feasibility will be assessed by:

- Percentage of patients eligible for the study who participated.
- Analysis of the reasons patients did not participate if they were eligible but not invited
- Percentage of eligible and invited patients who agreed to take part and consented
- Percentage of questionnaires fully completed (both questionnaires)
- Percentage of partially completed questionnaires (both questionnaires)
- Analysis of which questions were not answered where questionnaires are incomplete.
- Analysis of the percentage of patients completing just one questionnaire and reasons why
- Percentage of questionnaires not completed at all and reason why.

b) Acceptability will be assessed by:

- Analysis of the mean and range of scores given to the NICE based questionnaire for ease of use by radiographers
- Analysis of the mean and range of scores given to the NICE based questionnaire for ease of use by patients

A mean score of less than 5 would indicate the questionnaire was no more than moderately difficult to complete. Acceptability of use scores for both patients and staff will also be compared with clinical outcome, as a higher score may eventually be classed as acceptable should the questionnaire prove to have good predictive value.

- The mean and range of the time estimated that it took to complete the NICE questionnaire will be assessed.
- ii) Analysis of Secondary aims
- a) Results from the NICE based questionnaire will be compared to results from the BPI.
- b) Clinical data collected at the time of the scan will be compared with clinical outcome
- c) Questionnaire scores will be compared with clinical outcome

A comparison of the mean values and standard deviation of the scores from both questionnaires will be made. Relationships will be investigated using chi-square tests and logistic regression.

Ethical issues

Data collected for the purpose of the study will be compliant with the Data Protection Act, Caldicott principles and good research practice standards. No recruitment will commence until independent ethical, University and Trust R&D approval is granted.

Potential Anxiety/Distress

It could be argued that for some, the study title and the topic may cause unnecessary anxiety. The eligible patients are, however, already being managed for a serious disease or suspicion of cancer and therefore, questions about their symptoms would probably be expected by them. The patient information sheet and staff explanations will be explicit in reassuring patients that the questions are standardised for all patients and are not asked because of any individual clinical concern. The potential of creating unjustified anxiety for patients is, therefore, not expected to be problematic but patients will be cared for by experienced oncology radiographers, familiar with and sensitive to, discussing cancer with patients. As per normal practice, on site support from clinical nurse specialists and the Macmillan support team is available for any cases where additional counselling may be identified as required during the visit.

Management of High Risk Patients

It is not intended for the purpose of this study that any additional clinical action or intervention will take place for any patients but it is standard practice that staff advise patients to seek medical review if minor symptoms have changed since referral. When radiographers suspect a patient is displaying neurological symptoms which require urgent medical attention urgent referral to the on site oncology team is arranged.

Dissemination strategy

The results of the study will be shared within the Trust to the appropriate specialist groups, the Trust MSCC clinical lead and the regional MSCC working group. Submission for presentations will be made at the British Nuclear Medicine Society annual meeting and submission for publication in the radiographers' professional body journal.

References:

Bowen.D.J, Kreuter.M, Spring.B, Cofta-Woerpel.L, Linnan.L, Weiner.D, Bakken.S, Kaplan.C.P, Squiers.L, Fabrizio.C, Fernandez.M. 2009. How We Design Feasibility Studies.American Journal of Preventative Medicine. May 36(5): 452-457

British Nuclear Medicine Society. 2003. Clinical Guidelines – 99mTc Diphosphonate Bone Imaging for Metastases.

Buscholtz JC (1999) Metastatic Epidural Spinal Cord Compression. Seminars in Oncology Nursing. 15(3): 150-159

Cleeland. CS. The measurement of pain from metastatic bone disease: capturing the patients experience 2006. Clin Caner Res 2006;12(20 pt2):6236s-6242s.

Cleeland.CS, Portenoy RK, Rue M, Mendoza TR, Weller E, Payne R, Kirshner J, Atkins JN, Johnson PA, Marcus. Does an oral analgesic protocol improve pain control for patients with cancer ? An intergroup study coordinated by the eastern Cooperative Oncology Group. Ann Oncology 2005;16:972-80

Cleeland CS, Mendoza TR, Wang XS, Chou C, Harle MT, Morrisey M, Engstrom MC, Assessing symptom distress in cancer patients: The M.D.Anderson Symptom Inventory . Cancer 2000;89:1634-46

Cook AM, Lau TN, Tomlinson MJ (1998). Magnetic Resonace Imaging of the Whole Spine in Suspected Spinal Cord Compression Impact Management. Clinical Oncology (Royal College of Radiologists), 10:39-43

de Haes JC, van Knippenberg FC, Neijt JP. Measuring psychological and physical distress in cancer patients : structure and application of the Rotterdam Symptom Checklist. Br J cancer 1990;62:1034-8

Helweg-Larsen, S. (1996) Clinical outcome in metastatic spinal cord compression. A prospective study of 153 patients. Acta Neurologica Scandinavica, 94: 269–275.

Husband D.J, Grant K.A, Romaniuk C.S. (2001) MRI in the Diagnosis and Treatment of Suspected Malignant Spinal Cord Compression. British Journal of Radiology, 74: 15-23

Husband, D.J. (1998) Malignant Spinal Cord Compression: Prospective Study of Delays in Referral and Treatment. British Medical Journal, 317: 18-21

Husband, D. J. (1998) Malignant spinal cord compression: prospective study of delays in referral and treatment. BMJ, 317: 18–21.

International Atomic Energy Agency. 2007. Criteria for Palliation of Bone Metastases – Clinical Applications. IAEA-TECDOC-1549 April.

Levack, P, Graham J, Collie D, Grant R, Kidd J, Kunkler I, Gibson A, Hurman D, McMillan D, Rampling R, Slider L, Statham P, Summers D (2002). Don't Wait For Sensory Level – Listen to the Symptoms: A Prospective Audit of the Delays in Diagnosis of Maliganant Cord Compression. Clinical Oncology (Royal College of Radiologists), 14:472-480

Loblaw D.A, Laperriere N.J, Mackillop W.J. (2003). A Population Based Study of Malignant Spinal Cord Compression in Ontario. Clinical Oncology (Royal College of Radiologists). 15(4): 211-7

Maranzano, E. & Latini, P. (1995) Effectiveness of radiation therapy without surgery in metastatic spinal cord compression: Final results

from a prospective trial. International Journal of Radiation Oncology, Biology, Physics, 32: 959–967.

Mitera, G., Loblaw, A. (2003) Delays from symptom onset to treatment in malignant spinal cord compression: quantification and effect on pre-treatment neurological status. Radiother. Oncol 69: Abstract 141

Merseyside and Cheshire Cancer Network. 2012. Metastatic Spinal Cord Compression Pathway. Metastatic Spinal Cord Compression Working Group.

National Institute for Health and Clinical Excellence (2008). Metastatic Spinal Cord Compression: Diagnosis and Mangement of Patients at Risk of or with Metastatic Spinal Cord Compression. CG75

Serlin RC, Mendoza TR, Nakamura Y, Edwards KR, Cleeland CS. When is cancer pain mild, moderate or severe ?. Grading pain severity by it's interference with function. Pain 1995;61:277-84

Solberg, A. & Bremnes, R. M. (1999) Metastatic spinal cord compression: diagnostic delay, treatment, and outcome. Anticancer Research, 19: 677–684.

Turner S, M. B. T. I. B. J. (1993) Malignant spinal cord compression: a prospective evaluation. Int J Radiat Oncol Biol Phys, 26: 141–146.